Author’s response to reviews

Title: Does Pentaerythritoltetranitrate reduce fetal growth restriction in pregnancies complicated by uterine mal-perfusion? Study protocol of the PETN-study: a randomized controlled multicenter-trial

Authors:

Tanja Groten (tanja.groten@med.uni-jena.de)

Thomas Lehmann (thomas.lehmann@med.uni-jena.de)

Ekkehard Schleußner (ekkehard.schleussner@med.uni-jena.de)

Version: 2 Date: 11 Aug 2019

Author’s response to reviews:

Dear Editor, revisions were made as demanded. I would like to express my concerns about remaining text overlap. Please note, that some overlap, like explaining HO-1 is hemeoxygenase-1 and similar expressions are nearly not possible to avoid. May I ask to prove the necessity of further changes under this consideration?

Best regards Tanja Groten